Author's response to reviews

Title: Vitamin D and tuberculosis: a multicenter study in children.

Authors:

Elisabetta Venturini (elisabetta-venturini@virgilio.it)
Ludovica Facchini (facchini.ludovica@gmail.com)
Nuria Martinez-Alier (nuria.martinez-alier@gstt.nhs.uk)
Vas Novelli (Vas.Novelli@gosh.nhs.uk)
Luisa Galli (luisa.galli@unifi.it)
Maurizio de Martino (maurizio.demartino@unifi.it)
Elena Chiappini (elena.chiappini@unifi.it)

Version: 5 Date: 12 November 2014

Author's response to reviews: see over
Response to Editor:

"The final manuscript should have separate section on Ethical issues. Please follow the instructions to authors and it is a clear requirement that the section should stand alone."

A Ethical issues section has been added separately within the methods section.

Acknowledgements: By way of a section ‘Acknowledgements’, please acknowledge anyone who contributed towards the article by making substantial contributions to conception, design, acquisition of data, or analysis and interpretation of data, or who was involved in drafting the manuscript or revising it critically for important intellectual content, but who does not meet the criteria for authorship. Please also include the source(s) of funding for each author, and for the manuscript preparation. Authors must describe the role of the funding body, if any, in design, in the collection, analysis, and interpretation of data; in the writing of the manuscript; and in the decision to submit the manuscript for publication. Please also acknowledge anyone who contributed materials essential for the study. If a language editor has made significant revision of the manuscript, we recommend that you acknowledge the editor by name, where possible.

The role of a scientific (medical) writer must be included in the acknowledgements section, including their source(s) of funding. We suggest wording such as 'We thank Jane Doe who provided medical writing services on behalf of XYZ Pharmaceuticals Ltd.' Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgements section.

The Acknowledgement section was not added as all the authors made a substantial contribution, meeting the criteria for authorship. The language copyediting was done by VN and NMA.

Consent Statement: Please state in the Methods section whether written informed consent for participation in the study was obtained from participants or, where participants are children, a parent or guardian.

A consent statement has been added to the methods section, as suggested.

> Please update your ethics statement to include the name of the ethics committee that approved your study.

The name of the Ethical committee of the hospital involved has been included in the ethical section.

We recommend that you copyedit the paper to improve the style of written English.

The language copyediting was done by VN and NMA.
Please also ensure that your revised manuscript conforms to the journal style.

We formatted the manuscript according to the journal style.